Author's response to reviews

Title: Socioeconomic inequalities in smoking habits are still increasing in Italy

Authors:

Giuseppe Verlato (giuseppe.verlato@univr.it)
Simone Accordini (simone.accordini@univr.it)
Giang Nguyen (giang.nguyen@univr.it)
Pierpaolo Marchetti (pierpaolo.marchetti@univr.it)
Lucia Cazzoletti (czzlcu44@univr.it)
Marcello Ferrari (marcello.ferrari@univr.it)
Leonardo Anonicelli (lantonicelli@ospedaliriuniti.marche.it)
Francesco Attena (francesco.attena@unina2.it)
Valeria Bellisario (Valeria.bellisario@unto.it)
Roberto Bono (roberto.bono@unito.it)
Lamberto Brizarelli (lbrizigi@unipg.it)
Lucio Casali (lcasali@unipg.it)
Angelo Guido Corsico (corsico@unipv.it)
Alessandro Fois (Fois.Ale@libero.it)
Maria Grazia Panico (grazia.panico@tin.it)
Pavilio Piccioni (ppiccioni@qubisoft.it)
Pietro Pirina (pirina@uniss.it)
Simona Villani (svillani@unipv.it)
Gabriele Nicolini (G.Nicolini@chiesigroup.com)
Roberto de Marco (roberto.demarco@univr.it)

Version: 2 Date: 15 May 2014

Author's response to reviews:

we made the following changes:
- E-mails of 19 co-authors were added to the title page.
- Line numbers were added and page breaks were removed. Text was unjustified.
- “Introduction” header was changed to “Background”
- “Conclusions” header was added (line 319, page 14)
- In “tables and figures” page (page 21) we modified titles and legends of tables and figures to make them in line with the requirements of word count.